Author's response to reviews

Title: Charting the Progression of Disability in Parkinson Disease: Study Protocol for a Prospective Longitudinal Cohort Study.

Authors:

Leland E Dibble (Lee.Dibble@hsc.utah.edu)
James T Cavanaugh (jcavanaugh@une.edu)
Gammon M Earhart (earhartg@wusm.wustl.edu)
Terry D Ellis (tellis@bu.edu)
Matthew P Ford (mford@uab.edu)
Kenneth B Foreman (Bo.Foreman@hsc.utah.edu)

Version: 4 Date: 1 November 2010

Author's response to reviews:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is First name Initial Last name (e.g. Joe F Bloggs)

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Underlining: please remove.

Please see responses to editing requests below:

Tables- please remove the visible vertical lines from your tables: DONE

Figure: please list as Figure 1 in the text. DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. DONE

PLEASE LET US KNOW IF ANYTHING FURTHER IS NEEDED.

BEST REGARDS,

LEE DIBBLE AND CO-AUTHORS